
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Sun Aug  7 17:03:08 GMT 2005
Year: 05  Doy: 219
Observer: koon
WEATHER COMMENT: Sun Aug  7 17:03:10 GMT 2005
Rainy, no obs until weather clears.
___end___
**PSPT PROBLEM**: Sun Aug  7 17:33:46 GMT 2005
The data transfer tape was in the Kenobi tape drive this morning, instead of
the PSPT drive, I moved it to the PSPT drive.
___end___
A NEW TAPE HAS BEEN PUT INTO KAIEE DLT DRIVE, Sun Aug  7 17:41:24 GMT 2005
WEATHER COMMENT: Mon Aug  8 00:35:40 GMT 2005
The rain has finally stopped but the clouds are still thick and dark.  The
roof leaks are much smaller now - due to our efforts at preventing water
from coming down the inside of the dome cylinder wall.
___end___
Mon Aug  8 00:37:28 GMT 2005 
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